Author’s response to reviews

Title: Feasibility of Community Based HIV Self-Screening in South Africa: A Demonstration Project

Authors:

Limakatso Lebina (lebinal@phru.co.za)
Ntombexolo Seatlholo (Seatlholop@phru.co.za)
Noah Taruberekera (ntaruberekera@psi.org)
Mopo Radebe (mopo@sfh.co.za)
Anthony Kinghorn (kinghora@phru.co.za)
Tessa Meyer (tessa@sfh.co.za)
Miriam Mhazo (mariam@sfh.co.za)
Kennedy Otwombe (otwombek@phru.co.za)
Khuthadzo Hlongwane (hlongwanek@phru.co.za)
Ashley Ringane (ringanea@phru.co.za)
Nthabiseng Koloane (koloanen@phru.co.za)
Mbali Nkuta (www.nkutam@phru.co.za)
Nkhensani Nkhwashu (NkhensaniN@foundation.co.za)
Thato Farirai (ThatoF@foundation.co.za)
Patience Kweza (PatienceK@foundation.co.za)
Thato Chidarikire (www.matshabathato44@gmail.com)
Simukai Shamu (SimukaiS@foundation.co.za)
Tendesayi Kufa (TendesayiKC@nicd.ac.za)
Adrian Puren (adrianp@nicd.ac.za)
Neil Martinson (martinson@phru.co.za)
Dear Editors

Re: Manuscript: "Feasibility of HIV Self-Screening in South Africa: A mixed methods study" - PUBH-D-18-02534

Thank you very much for taking the time to review our manuscript entitled Feasibility of HIV Self-Screening in South Africa: A mixed methods study.

Please see below the responses for each of the comments.

Editors Comments

1. Authors’ contributions

   Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

   Thank you this comment, the authors contribution has been updated for clarity. (Line 351-357)

2. Consent to participate

   In this section of the Declarations, please clarify whether ALL participants provided consent to participate before commencement of the study.
Thank you this comment, we have included the sentence that all participants provided informed consent prior to participation. (Line 336-337)

3. Please ensure that all figures and tables are cited in the main body of text. Please note that any that aren’t cited may be deleted upon production of the manuscript.

Thank you this comment, we have ensured that all figures and tables are referenced in the text. (Lines 186; 216; 229; 241)

Thank you for reviewing the manuscript.

Yours Sincerely

Limakatso Lebina and the co-authors.